Author's response to reviews

Title: Neuroprotective effects of Psoralea corylifolia Linn seed extracts on mitochondrial dysfunction induced by 3-nitropropionic acid

Authors:

A-Rang Im (lar744@hanmail.net)
Sung-Wook Chae (kendall@kiom.re.kr)
Gui jun Zhang (guijunzhan@163.com)
Mi-Young Lee (lar747@kiom.re.kr)

Version: 4 Date: 12 September 2014

Author's response to reviews: see over
To dear Prof. Editor.

I have received the letter from you asking to resubmit after change of our manuscript, according to the comments on September 11, 2014.
Thank you very much for your reconsideration to give us a chance to change our manuscript. As you advised, we properly revised the manuscript. I hope that our work is suitable for the publication of *BMC Complementary and Alternative Medicine* and I’m looking forward to favorable response from you soon.

**Comments by Reviewer #1:**
Reviewer's report:
Authors have addressed almost all the issue I raised out from the initial manuscript. I think that the article can be published at its present state.

**Comments by Reviewer #2:**
Reviewer's report:
Additional information is not provided particularly on ingredient identification and its effect. The authors told that the comment I made they want to do in future experiments. It is up to discretion of Editor to accept or reject.

**Editorial comments:**
We must again request that you provide details in your Methods section on who undertook the formal identification of the plant material used in your study. Please also confirm whether a voucher specimen of this material has been deposited in a publicly available herbarium, and include this information in your manuscript. A deposition number should be included, if available.

*Answer:* *P. corylifolia* seed was purchased from Kwangmyungdang Medicinal herbs (Ulsan, Korea). A voucher specimen (KIOM111930, KIOM211930) has been deposited at the herbarium of the Department of Aging Research Lab., Korea Institute of Oriental
Thank you for your consideration. I look forward to hearing from you.

Sincerely,
Dr. Mi Young Lee
KM-Based Herbal Drug Development Group,
Korea Institute of Oriental Medicine
1672 Yuseongdae-ro, Yuseong-gu, Daejeon, 305-811, Korea
Tel: +82-42-868-9504; Fax: +82-42-863-9301